Author's response to reviews

Title: Antiproliferative activity and induction of apoptosis by Annona muricata (Annonaceae) extract on human cancer cells

Authors:

Constant A Pieme (apieme@yahoo.fr)
Santosh G Kumar (santoshkumarguru@gmail.com)
Mireille SG Nguepi (syldong@yahoo.fr)
Moukette M Bruno (moukettebruno@gmail.com)
Fabrice F Boyom (fabrice.boyom@fulbrightmail.org)
Jeanne Y Ngogang (ingogang@yahoo.fr)
Ajit K Saxena (aksaxena@iiim.ac.in)

Version: 2
Date: 5 June 2014

Author's response to reviews: see over
To whom it may concerned

Object: Submission of the manuscript

Dear Sir

I, the undersigned corresponding author of the manuscript titled “Antiproliferative activity and induction of apoptosis by Annona muricata (Annonaceae) extract on human cancer cells” hereby declare that the manuscript which is submitted for publication in the Environmental Toxicology and Pharmacology is not published elsewhere.

This manuscript described the cytotoxic effect and the mechanism of apoptosis exert by extracts from leaves, roots and stem barks of Annona muricata. No scientific study has been carried on this purpose.

I hope that this study which is original will be published into your journal.

Looking forward to hearing from you.

Yours sincerely

Pieme Constant Anatole

Department of Biochemistry and Physiological Sciences; Faculty of Medicine and Biomedical Sciences, University of Yaoundé I, PO Box 1364 Yaoundé, Cameroon.

Email: apieme@yahoo.fr Phone: 00237 74 55 18 71